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KINGSTON ASH RECOVERY PROJECT

WEEKLY REPORT
May 25 - 31, 2009

HIGHLIGHTS

Recovered just under 20,000 cubic yards of ash from the river.

TVA 14" dredge is undergoing preparation for movement to the river.

Track #2 (rail spur) is complete up to the guard shack.

Best and final offers received and evaluated for loading and disposal.

Tree removal and hole repair on Dike C completed. Stability analysis results for Dike C presented with a
safety factor of 1.45.

INFRASTRUCTURE/ASH MANAGEMENT

Engineering

e Car wash design and installation still on hold — determination of location is pending.

e Transmitted the elevation survey and layout results to CTI (Utilities Engineer) so they can complete the
engineering of the enhancement for Kingston water line.

e Prepared sketch to widen the radius at the intersection of Swan Pond Road and Swan Pond Circle. Task
Order to be issued to the Civil Projects Group.

e Preparing sketch to eliminate the S curve in the middle entrance road to the facility.

Roads
e The ramp at Dike C is complete and operational. With the ramp complete, we have implemented our new
traffic flow plan making traffic one way on Dike C.

Utilities
e No construction activity at this time.

Water Management

¢ Installing the stand pipes and preparing to install the fabric baffles in Sediment Pond 2A. Also installed the
discharge lines from 2A to the clean water ditch on the east end of the pond.

¢ Installed the 3 — 48" discharge pipes through the road from the east embayment.

e Have hauled approximately 76,850 cubic yards of ash from the Sediment Ponds and area east of Dike 2 to
the Ash Processing Area.

e Continued construction of the clean water ditches — from south of Swan Pond Circle to Dike 2.

e Opverall construction of the water drainage impacted by the rain received during the week.

e Completed installing the temporary rock road in the north embayment for accessibility to construct the
berm and move the ash from the northern most point of the embayment. Work under the power lines in
the north embayment is scheduled for next week. A TVA lineman will be on site during this work.

e Continued installation of check dams and re-contouring of area for effective water management/drainage.

e Performed Storm Water Pollution Prevention Plan (SWPPP) inspections as required.

Dust/Erosion Control

o No Flexterra applied this week — contractor did not work because of the rain received during the week.

e 321 acres of Flexterra applied since January 31, 2009.

¢ Continuing operating water trucks on roads as needed for dust control.

o  Working with TVA personnel on additional short and long term remedies for dust control — seeding,
alternatives to Flexterra.

e Revised the Fugitive Dust Control and Suppression for submittal to the EPA.

ASH DREDGING & PROCESSING

Environmental Data Collection

e Continued daily TSS monitoring at KIF NPDES 001. The May monthly average through Thursday May 21
is 10.23 mg/L.

e Dredge plume monitoring was performed Tuesday through Saturday, May 26-May 30.
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Dredging

River dredging recovered approximately 19,500 cubic yards between May 24" and May 31%.

Dredging cumulative to date volume: 176,000 cubic yards.

At 1330 on May 29, 2009, dredge Emory released less than 1 gallon of biodegradable hydraulic fluid from
the cutterhead into the Emory River. The seal on the cutterhead was damaged by river debris (strong
vine) during dredging activities.

At approximately 1815 on May 29, 2009, dredge Luzon released less than 5 gallons of biodegradable
hydraulic fluid from the ladder winch into the Emory River near the KIF intake channel.

Both releases were contained and cleaned. The Luzon spill required assistance from the site spill
response contractor.

Ash Stilling Pond dredging continued after the Memorial Day holiday and approximately 2,000 cubic yards
were recovered this week.

A working pad was established near the Ash Settling Pond and a small crane was set up on the pad. The
crane was used to remove the ladder, cutterhead, spuds, and other items while disassembling the TVA
dredge in preparation for deployment in the river.

Crews performed maintenance work on the TVA dredge including removing petroleum based fluids and
replacing filters.

Prepared a work plan for more aggressive debris removal in Segment 1 and submitted the plan to EPA for
approval.

Ash Processing

The Ash Processing and Storage area contains approximately 260,000 cubic yards of fly ash. A portion of
the stockpile is from dredging and recovery operations and the remainder is from the embayment area
(material excavated from the installation of stormwater ponds and clean/dirty water ditches) and
subsurface improvements in support of rail spur installation adjacent to the Ash Processing and Storage
Area.

ASH DISPOSITION

Construction on the rail spur is ongoing. Track # 2 is complete to station 5+00, which brings it down to the
area adjacent to the guard shack. The plant entry road has been relocated south into the existing parking
area, and drainage pipe has been installed to accommodate the rail turnout off the Norfolk Southern main
track. Site prep for Track #1 continues. The project is on schedule for June 16™ completion of Track #2.
Unloading of the test rail cars that did not leave the KIF site continues. This activity is scheduled to finish
June 1% or 2.

Best and final offers for the ash loading and disposal contractors were received and evaluated. Letters of
intent should be issued early the week of June 1°'.

TransAsh was given the o.k. to cover the remaining conduit for the Geosyntec piezometer and settlement
plates with processed ash. Geosyntec online instrumentation readings will be available the week of June
1% after IT issues are resolved.

A plan to protect slope inclinometer #3 (SI-3) was established with Phillips and Jordan during shot rock
installation. The shot rock was installed successfully around SI-3, and it appears that it should remain
functional.

CENOSPHERE RECOVERY

Completed removal of heavy debris from the river with the excavator.

Completed removing TVA acquired boats. Boats are being stored at the staging area.

Preparing to demobilize two of the six VTU barges. VTUs to be removed on Monday and barges to be
removed later in the week. Coordinating the lifts with CPG.

Worked on area known as the Vista Estates area. This is the last of the impacted areas directly affecting
residents. There are several shoreline calls that are still ongoing.

SKIMMER WALL

The decision to suspend dredging in the Skimmer Wall area was made on Friday May 29. This decision
was made due to concerns with elevated turbidity levels that may occur in the intake channel during peak
power demand of the summer.

Dredging to uncover disabled wall will commence during October outage at KIF.

Estimated installation date of new Skimmer Wall will begin after October outage and will be complete
before summer peak period of 2010.
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e Turbidity control plan measures will be analyzed and implemented in the short term to provide protection
to the plant intake until the new Skimmer Wall is installed.

DIKE REINFORCEMENT

e Stantec has completed 57 of the 63 proposed borings. The six remaining proposed borings require a
barge mounted drill rig. Stantec is awaiting an increase in funding before proceeding with mobilizing a
crane and barges.

¢ Stantec made a presentation on the dike stability to the regulators. The stability of Dike C is approximately
1.45.

¢ Completed removing the trees and backing holes with dense grade aggregate along Dike C. Repairing low
areas with new riprap will begin on Monday.

ROUTINE MONITORING

Surface Water Sampling

¢ Routine surface water fixed station monitoring occurred on Wednesday, and Friday, May 27 and 29. On
Tuesday May 26, monitoring at the fixed stations occurred due to a >0.5” rain event.

e The monthly sampling and toxicity testing of the KIF Stilling Pond discharge continued through Friday May

29.
Air Sampling

e Fixed-location and real-time monitoring continued. No exceedences of standards were observed from
real-time measurements.

Data Management

¢ Electronic data packages continue to be received from the metals re-analysis of surface water samples
collected in January and February 2009.

e Data validation continues for electronic data packages received from the laboratories. As they are
received, the verified packages are being posted to the password protected website for regulator review.

Other Projects

o The split sample analyses data from the Pace Laboratories of fish fillets collected by TWRA, TDEC, TVA,
and ORNL were posted on the public web site. TDEC has also posted the results from their analyses of
their splits. Data have yet to be received for splits analyzed at the TVA central laboratory and an ORNL
contract laboratory.

FINAL REMEDIATION

Engineering
e Engineering and Project Management discussed TDEC issues on the dredge cell embankment test. The
project is working on a solution to the issues raised that will allow approval of the work plan.

Documentation

e Work continues on preparing for the upcoming data quality objectives workshop. Preliminary human
health and ecological conceptual site models were developed. Tables screening site related data against
appropriate risk benchmarks have been developed.

SAFETY

There were 11 site specific orientations performed during the week - 291 to date.

There were 2 Safety Observation Reports written, 21 of which have been closed.

The first 40-hour HAZWOPER training class was completed for 23 attendees.

There were 3 incidents involving releases of hydraulic fluid:

At 1330 on May 29, 2009, dredge Emory released less than 1 gallon of Chevron Clarity Hydraulic oil from

the cutterhead into the Emory River. The source of the release appeared to be from the seal in the

cutterhead shaft.

e At 1745 on May 29, 2009, dredge Luzon released approximately 5 gallons of Chevron Clarity Hydraulic
into the Emory River. The source of the release appeared to be from the main winch.

e At 1605 on May 30, 2009, water truck number WT-43 released approximately 5-7 gallons of Rotella T

Engine Oil SAE 10@30 Hydraulic onto the roadbed of the hill and C-Dike roadbed just north of the bridge
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over the Plant intake channel. The source of the release appeared to be from a ruptured hydraulic hose in
the engine compartment.

COMMUNICATIONS

Communications

e Developing communications strategy and materials for Root Cause Analysis.

e Developing communications strategy and materials for selection of offsite disposal contractors.

o Posted to TVA’s Kingston Webpage: fish testing results, Swan Pond Road lane closures, Updates on
Water and Air Sampling.

Outreach Events

Met with local residents to gather ideas for next open house/public meeting.

Continued community interviews in preparation for the CIP (40+ interviews completed).

Mailed letters to residents who may be impacted by changes in the flood plain due to the ash spill.
Continued appraisals, offers and closings on properties TVA is willing to purchase.

Responded to numerous Congressional Inquiries and TVAINFO letters about residents.

Fielded calls and visits from residents about various issues.

RIVER OPERATIONS

Weekly Average &
Elevations and Flows d@‘

Q)
June 01, 2009 e
(7 day average of data through May 31) N
Emory River
Kingston Fossil Plant Flow: 454 cfs
$-\.‘ef

Cy;,

W Melton Hil

Flow: 3,125 cfs
Highway 58 Elev: 794.03 ft
Bridge

Average Kingston Velocity
0.424 ft/sec

Fort Loudoun
Flow: 18,721 cfs
Elev: 812.97 ft

Watts Bar
Flow: 22,670 cfs
Elev: 740.87 ft
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MEETINGS/CONFERENCE CALLS

Date | Time Subject

06/01 | 1500 EPA/TDEC/TVA Telecon

06/02 | 1000 TCM Training — ICP Conference Room

06/02 | 1300 Document Control Meeting — ICP Conference Room
06/03 | 0900 KIF Recovery Staff Meeting — ICP Conference Room
06/03 | 1300 Construction Alignment Meeting — ICP Conference Room
06/04 | 1400 Rail Spur Construction Meeting — ICP Conference Room
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Kingston Recovery Project

Date:

May 25-31, 2009

Field Report

Safety Status/Personnel Casualty Summary

Since Last Report Adjustments To Previous Op Period Total Involved
First Aid 0 0 9
Recordable Incidents 0 0 2
Lost/Restricted Time Incident 0 0 0
Cenosphere/Debris Removal Quantity
Recovered Materials Weekly Total Cumulative Total
Outreach Light Debris Removal — Land (bags) 0 18,867
Divisions A, B,C,D Cleanup Division shoreline cleanup (bags) 1,842 60,449
Total Tons of Light Debris 0 869.69
Cenosphere Slurry (gallons) 150,000 5,386,500

Environmental

. Weekly Samples Total # of Samples Sample results
Org Matrix CoBI/Iectedp Collected to dgte received verified
TVA Air (Real-Time)* 2,379 46,911 46,911 46,911
TVA Air (Analytical)* 19 1,405 1,277 1,039
TVA Surface/Utility Water 62 1,469 1,345 198
TVA Ground Water (spring and well) 0 5 5 2
TVA Ash 0 81 86 50
TVA Soil/Sediment 13 76 122 9
i Cumulative

Dredging Weekly Total Total
River Estimated Cubic Yards 19,500 176,000
Ash Pond 0 45,000
Stilling Pond Estimated Cubic Yards 2,000 5,100
Comments:
*May 22-28, 2009
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